Author’s response to reviews

Title: Prevalence of cervical infection with HPV type 16 and 18 in Vietnam: Implication for vaccine campaign

Authors:

Lan T. H. Vu (vhl@hsph.edu.vn)
Dieu Bui (lth3@hsph.edu.vn)
Ha T. T Le (hale_hsph@yahoo.com)

Version: 2 Date: 30 September 2012

Author’s response to reviews:

Dear Editor,
I had made the following changes as you requested:
- Including a title page following the suggested format
- Formatting the abstract according to the guidelines for authors
- Adding a statement about no competing interest (page 8)
- Adding a section about author’s contribution (page 8)
Thank you very much.
Lan Vu